Author's response to reviews

Title: Cumulative exposure to air pollution and long term outcomes after first acute myocardial infarction: A population-based cohort study. Objectives and methodology

Authors:

Yariv Gerber (yarivg@post.tau.ac.il)
Vicki Myers (vicki_myers@hotmail.com)
David M Broday (dbroday@techunix.technion.ac.il)
Silvia Koton (koton@post.tau.ac.il)
David M Steinberg (dms@post.tau.ac.il)
Yaacov Drory (drory.yaacov@gmail.com)

Version: 4 Date: 24 June 2010

Author's response to reviews:

The current submitted version includes several small typographical corrections, text clarifications and rewritings.